Author's response to reviews

Title: Case-based exercises fail to improve medical students' information management skills: a controlled trial

Authors:

Heidi S. Chumley (hchumley@kumc.edu)
Alison E. Dobbie (alison.dobbie@utsouthwestern.edu)
John E. Delzell Jr. (jdelzell@kumc.edu)

Version: 4 Date: 17 February 2006

Author's response to reviews:

Formatting change: Renamed figure 1 as table 1. As such table 1 became table 2 and table 2 became table 3.